Author’s response to reviews

Title: Building school-based cardiovascular health promotion capacity in youth: a mixed methods study

Authors:

Roberta L Woodgate (Roberta.Woodgate@umanitoba.ca)
Corey M Sigurdson (Csigurds9@myumanitoba.ca)

Version: 6 Date: 16 April 2015

Author’s response to reviews: see over
TO: Receiving Editor, BMC Public Health  
FROM: Roberta Woodgate  
DATE: April 16, 2015  

SUBJECT: MANUSCRIPT SUBMISSION

I am resubmitting the manuscript entitled *Building school-based cardiovascular health promotion capacity in youth: a mixed methods study* to *BMC Public Health* for publication. It reports on a mixed methods study conducted to determine if providing a research intervention in the form of education, empowerment, and support builds youth’s capacity for cardiovascular health promotion. This study is innovative as this approach to youth cardiovascular health promotion not only provides youth with knowledge of cardiovascular health, it fosters the development of life skills such as leadership and advocacy that support long-term health.

Please see the following page for requested revisions.

*This manuscript presents original, unpublished material and is not under consideration for publication elsewhere. There are no conflicts of interest associated with this manuscript. Both authors have approved this manuscript for submission.*

If you have any questions, please contact me at: robeerta.woodgate@umanitoba.ca or (204) 474-8388 and thank you kindly for considering our work.

Sincerely,

Dr. Roberta L. Woodgate  
Canadian Institutes of Health Research Applied Chair in Reproductive, Child and Youth Health Services and Policy Research  

Professor, Child Health and Illness  
Faculty of Health Sciences, College of Nursing  
University of Manitoba  
Winnipeg, Manitoba  
Canada, R3T 2N2  
Phone: 204-474-8338; Email: Roberta.Woodgate@umanitoba.ca
There was only one revision requested.

<table>
<thead>
<tr>
<th>Recommendations</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Please update your ethics statement to include the name of the ethics committee that approved your study.</td>
<td>On page 11 (lines 227-229) we added our ethics committee name.</td>
</tr>
</tbody>
</table>